Author's response to reviews

Title: MMP-9 expression varies according to molecular subtypes of breast cancer

Authors:

Einas M Yousef (enasesawy@hotmail.com)
Muhammad R Tahir (ramzantahir7@gmail.com)
Yves St-Pierre Prof. (yves.st-pierre@iaf.inrs.ca)
Louis A Gaboury Prof. (louis.gaboury@umontreal.ca)

Version: 3 Date: 9 April 2014

Author's response to reviews: see over
Dear Dr Yousef,

Thank you for submitting your manuscript to Breast Cancer Research. The Editorial Board has carefully considered the information presented in your manuscript and we regret to inform you that your above-referenced submission will not be acceptable for publication. We are therefore returning your manuscript so that the work may be considered elsewhere without further delay.

Breast Cancer Research is receiving an increasing number of submissions, and manuscripts accepted for publication must therefore meet stringent considerations of scientific impact, breadth, quality, and originality. They must also be judged to be of significant general interest to the audience of the journal. Unfortunately, in this instance it was felt that these criteria were unlikely to be fulfilled to the extent needed to compete successfully for priority relative to other submissions under consideration.

I am sorry that on this occasion we could not render a more favourable decision on your manuscript. However, we truly appreciate the opportunity to review your work and we would welcome future submissions from your group.

We would like to offer you the opportunity to transfer your manuscript, with any required revisions, to an alternative journal within the BioMed Central portfolio: we suggest BMC Cancer [http://www.biomedcentral.com/bmccancer/about].

Please note, BMC Cancer is editorially independent from Breast Cancer Research and the transfer of your manuscript does not imply that it will be accepted by the journal. The Editors of BMC Cancer are free to conduct their own peer review or reject the manuscript if it is not suitable.

Benefits of transferring your submission of this manuscript to BMC
Cancer may include:

* easier publication of your work, saving time finding and submitting to an alternative publisher;
* faster publication, we will transfer your manuscript record and reviewer comments to the suggested journal for you;
* reaching the right audience for your work, we believe the suggested journal may be a better fit.

Your submission record and files, including any reviewers' comments, will be transferred and you are able to complete submission to BMC Cancer by logging on to 'My Manuscripts' [http://www.breast-cancer-research.com/my/manuscripts].

Please note, BMC Cancer operates using a different peer-review system to Breast Cancer Research. It uses an open peer-review system. Open peer review means that, firstly, the reviewers' names are included on the peer review reports, and secondly that, if the manuscript is published, the reports are made available online along with the final version of the manuscript. The published article will provide a link to its 'pre-publication history', which lists all the versions of the manuscript, all the signed reviews, and all responses to the reviewers since the submission of the manuscript until its publication. Please refer to http://www.biomedcentral.com/bmccancer/about#publication for further details on the peer-review policy of BMC Cancer, or if you have any questions, please contact editorial@biomedcentral.com

With all best wishes,

The Breast Cancer Research Editorial Team
Email: editorial@breast-cancer-research.com